

November 27, 2020

Prof. Lian-Sheng Ma
Editor-in-Chief
World Journal of Clinical Cases

Dear Prof. Lian-Sheng Ma,

Many thanks for the letter dated November 20, 2020, regarding our manuscript entitled "*Congenital hepatic fibrosis in a young boy with congenital hypothyroidism: a case report*" (Manuscript NO: 60566). We appreciate the valuable comments of the editor and reviewers. Based on the editor and reviewer's comments and suggestions, we have revised our manuscript. Our specific point-by-point responses to the editor and reviewers' comments and suggestions are as follows:

Reviewer 1:

1. According to case report, MRI and CT scan were done for case. Please discuss why MRI and CT angiography were done.

Response: We thank the reviewer 1 for the valuable comments. We have revised the discussion and added the reason why MRI and CT angiography were performed for this patient (see Page 8, second paragraph).

2. Please describe about role of the imaging procedures in discussion.

Response: We appreciate the reviewer 1 for the valuable comments. This has been added in the revised manuscript (see Page 9, first paragraph).

3. Please check line 157-158 "The compound glycyrrhizin tablet was discontinued at the 15th month when the patient was discharged from the hospital. " I think 15 months after hospital discharge is more better.

Response: We fully agree with the valuable suggestion. The sentence you mentioned has been rewritten in the revised manuscript (see Page 11, second paragraph).

4. Please check the Figure 3 that mentioned" and the compound glycyrrhizin tablet was discontinued at the 15th months post discharge"

Response: Thanks very much for your suggestion. The sentence has been revised (see page , second paragraph).

We hope that our revised manuscript is now suitable for publication in ***World Journal of Clinical Cases.***

Sincerely Yours,

Ting Zhang, MD, PHD,
Department of Gastroenterology, Hepatology, and Nutrition, Shanghai Children's
Hospital
Shanghai Jiao Tong University
Email: zhanqt@shchildren.com.cn